Author's response to reviews

**Title:** Self-Management Experiences among Men and Women with Type 2 Diabetes Mellitus: A Qualitative Analysis

**Authors:**

Rebecca T Mathew (rmathew@nosm.ca)
Enza Gucciardi (egucciari@ryerson.ca)
Margaret DeMelo (margaret.demelo@uhn.on.ca)
Paula Barata (pbarata@uoguelph.ca)

**Version:** 3 **Date:** 2 December 2012

**Author's response to reviews:** see over
To the Editorial Team at BMC Family Practice:

Thank you for your consideration of our article titled, “Self-Management Experiences among Men and Women with Type 2 Diabetes Mellitus: A Qualitative Analysis” (MS: 1398343811730763).

Please find below summary of the editorial changes requested by the BMC Family Practice Editorial Team.

1. Can you please name in your manuscript the specific ethics that approved your study?

Please see page 4, lines 8-9 (underlined text) for this information:

“The ethics approval for this research study was obtained from the University Health Network Research Ethics Board in Ontario, Canada.”

2. Can you please restore the title page and the competing interests, author contributions and acknowledgements section that were present in your initial manuscript but which are not present in the revised version?

Please see page 1 (title page) and page 15, lines 3-14 (underlined text) for the competing interests, author contributions and acknowledgements section.

Thank you for re-considering our paper for publication. Please do not hesitate to contact us if there are any further questions or concerns.

Rebecca Mathew, MD, HBSc
Physician Resident
Department of Internal Medicine
McMaster University
Hamilton, Ontario